Author’s response to reviews

Title: A Novel Germline ARMC5 Mutation in a Patient with Bilateral Macronodular Adrenal Hyperplasia: a case report

Authors:
Qiuli Liu (liuqiuli900827@163.com)
Dali Tong (tongdali1985@163.com)
Jing Xu (jingxu8023@163.com)
Xingxia Yang (yang_xing_xia@163.com)
Yuting Yi (yiyt@geneplus.org.cn)
Dianzheng Zhang (dianzhengzh@pcom.edu)
Luofu Wang (wangluofu@aliyun.com)
Jun Zhang (zhjunkk@163.com)
Yao Zhang (zy921-1@163.com)
Yaoming Li (finleylym@126.com)
Lianpeng Chang (changlp@geneplus.org.cn)
Rongrong Chen (chenrr@geneplus.org.cn)
Yanfang Guan (guanyf@geneplus.org.cn)
Xin Yi (yix@geneplus.org.cn)
Jun Jiang (jiangjun_64@163.com)

Version: 3 Date: 09 Mar 2018

Author's response to reviews:

Dear editor:

In response to the editor’s comments to the manuscript (Paper #: MGTC-D-18-00001R2) titled “A Novel Germline ARMC5 Mutation in a Patient with Bilateral Macronodular Adrenal Hyperplasia: a case report”, we have made the revision of the manuscript with the following itemized and point-by-point response to editor’s comments.
Comments

Your manuscript "A Novel Germline ARMC5 Mutation in a Patient with Bilateral Macronodular Adrenal Hyperplasia: a case report" (MGTC-D-18-00001R2) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Medical Genetics, once you have carried out some essential revisions:

Answer: Many thanks to your comments.

1. Please amend your Availability of data and materials statement, as your manuscript is a Case Report, please change the statements to "All data generated or analysed during this study are included in this published article".

Answer: We have changed the statements as you suggested (Page 12, line 9).

2. Please remove any supplementary files which do not want published, including the CARE checklist, as they are no longer required. Please include a statement in the Case presentation section to the effect that the CARE guidelines were followed when reporting this case.

Answer: We have removed the unnecessary supplementary files and included the statement in the Case presentation section to the effect that the CARE guidelines were followed when reporting this case (Page 8, line 20).

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: We have uploaded our manuscript as a single, final, clean version. And all the figures and tables have been uploaded as suggested.